Author's response to reviews

Title: No increased risk of colon cancer associated with various HRT formulations: A case-control study.

Authors:

Juergen C Dinger (dinger@zeg-berlin.de)
Lothar AJ Heinemann (heinemann@zeg-berlin.de)
Sabine Moehner (moehner@zeg-berlin.de)
Thai DoMinh (dominhthai@zeg-berlin.de)
Anita Assmann (assmann@zeg-berlin.de)

Version: 2 Date: 8 December 2006

Author's response to reviews:

We found a copy mistake in the submitted manuscript and re-submit the correct version. Sorry for any inconvenience created.
Lothar A.J. Heinemann
Dec. 7, 2006